Dr. Christian Scheideler Baltimore, Oct. 23 2002 Johns Hopkins University

Theory of Network Communication Fall 2002 Assignment 6

Problem 15 (4 points):

Try to come up with an example of a point distribution V so that GG(V ), Y G_Θ(V ), SpY G_Θ(V ), and SyY G_Θ(V ) all look different, where Θ = π/3.

Problem 16 (4 points):

Give an example that shows that the symmetric Yao graph is not a c-spanner for some constant c. (Hint: use two sets of nodes along two lines that are slightly non-parallel to each other to prevent the symmetric case from occurring for many node pairs.) How would the graph look like when using the sparsified Yao graph on this point set?